Author's response to reviews

Title: Interpersonal Behaviors and Socioemotional Interaction of Medical Students in a Virtual Clinical Encounter

Authors:

Olivier Courteille (olivier.courteille@ki.se)
Anna Josephson (anna.josephson@ki.se)
Lars-Olof Larsson (lars-olof.larsson@vgregion.se)

Version: 3 Date: 16 February 2014

Author's response to reviews:

Subject: MS: 1363839148108979

Manuscript: Interpersonal Behaviors and Socioemotional Interaction of Medical Students in a Virtual Clinical Encounter

We agree to the recommendation by the associate editor and we have tried to clarify the alignment between the posted Aims/Research questions and the described results. Indeed, inspired by the previous review, we posted too many Research questions and we are now more restrictive and some of the questions have consequently been taken away.

The manuscript has been sent to language proofing and is linguistically revised by an authorized expert.

Sincerely yours,

Lars-Olof Larsson
Corresponding author